Author’s response to reviews

Title: Evaluation of antioxidant activity and cytotoxicity of extract and fractions of Nardostachys jatamansi DC in breast carcinoma

Authors:

Shilpee Chaudhary (shilcdy@gmail.com)
Kodangala S Chandrashekar (cksbhat@yahoo.co.in)
Karkala S R Pai (ksr.pai@manipal.edu)
Managanahalli M Setty (mm.setty@manipal.edu)
Raviraj A Devkar (ravirajdevkar@gmail.com)
Neetinkumar D Reddy (reddy.neetinkumar95@gmail.com)
Shoja M Haneefa (shojamh@gmail.com)

Version: 2 Date: 21 August 2014

Author’s response to reviews: see over
August 21, 2014

From,
Dr. K S Chandrashekar,
Associate Professor,
Department of Pharmacognosy,
Manipal College of Pharmaceutical Sciences,
Manipal University,
Manipal-576104,
Karnataka, India.

To,
The Editor,
BMC Complementary and Alternative Medicine.

Dear Sir,

Subject: Re-submission of manuscript for evaluation.

Thank you for your comments. I am sending you herewith our manuscript titled, “Evaluation of antioxidant activity and cytotoxicity of extract and fractions of Nardostachys jatamansi DC in breast carcinoma”. We have made the required changes in the manuscript. The changes made are mentioned below:

- The line numbers have been included in the manuscript.
- The page numbers are included in the manuscript.
- No page breaks are there in the manuscript.

We believe that the manuscript is suitable for further review.

Thank you.

Sincerely,
K. S. Chandrashekar